December 2021

Dear Reviewer,

Thank you for your review and comments on “Assessing the Academic Achievement of US Orthopaedic Departments” for consideration of publication in the World Journal of Orthopedics.

In response to your questions, please see as follows.

1. There were no instances in which an orthopaedic department was home to more than one orthopaedic residency program. While some of these programs work together or share partnerships, faculty in this scoring system were considered in regard to his or her primary departmental home. In this way, faculty were not double counted, and the resultant scores reflect only one singular program.

2. The Hirschprung index (h-index) was used in this study as it is currently the best means of accurately reflecting the scholarly impact of authors’ publications. By including this in the scoring algorithm, equal weight was given to both the number of publications and quality of publications.

3. This algorithm does not reflect the quality of a training program, as is emphasized in the article discussion, given that high-quality surgical technique and management of orthopaedic conditions can be both taught and learned regardless of publication status and scholarly contribution.

4. It is an excellent point that many factors of the proposed scoring system apply solely to programs within the United States. Further adaptations to include equivalent or alternative funding sources and internationally representative leadership societies would be required to adequately assess programs outside the United States. This may be one possibility for further directions of this study.

We thank you for your questions and contributions to this study.

Sincerely,

Nicholas M. Bernthal, M.D.
Chief, Orthopaedic Oncology
UCLA / ORTHOPAEDIC INSTITUTE FOR CHILDREN
DEPARTMENT OF ORTHOPAEDIC SURGERY
DAVID GEFFEN SCHOOL OF MEDICINE AT UCLA
1250 16th Street, Suite 3142
Santa Monica, CA 90404
www.ortho.ucla.edu

UCLA/Orthopaedic Institute for Children Department of Orthopaedic Surgery
UCLA-Santa Monica Clinic • 1250 16th Street, Suite 2100 • Santa Monica, CA 90404
UCLA-Spine Clinic • 1131 Wilshire Boulevard, Suite 100 • Santa Monica, CA 90401
UCLA-Westwood Clinic • 100 UCLA Medical Plaza, Suite 755 • Los Angeles, CA 90095
Page Operator: (310) 206-6766

Francis J. Hornicek, MD, PhD
Professor and Chair
Anthony Scaduto, MD
Vice Chair, CEO, OIC
David McAllister, MD
Vice Chair of Academic Affairs
Susan Bukata, MD
Vice Chair of Clinical
John Adams, MD
Vice Chair of Research
Don Y. Park, MD
Vice Chair of Quality and Innovation
Nelson SooHoo, MD
Residency Program Director
Richard Bowen, MD
Assoc. Residency Program Dir.
Sharon Hame, MD
Med Student Edu. Program Dir.
Peter Alexakis, MD
Hand Surgery
Kodi Azari, MD
Prosper Benhaim, MD*
Neil Jones, MD
John Lawrence, MD
Joint Replacement
Edward J. McPherson, MD
Adam A. Sassoon, MD, MS
Alexandra Stavrakis, MD
Bert Thomas, MD
Erik N. Zeegen, MD*
Musculoskeletal Oncology
Nicholas Bernthal, MD*
Susan Bukata, MD
Brooke Crawford, MD
Francis J. Hornicek, MD, PhD
Pediatric Orthopaedics
Orthopaedic Institute for Children
Jennifer Beck, MD
Richard E. Bowen, MD
James Luck, MD
William L. Oppenheim, MD
Anthony Scaduto, MD*
Mauricio Silva, MD
Rachel M. Thompson, MD
Spine Surgery
Allen Chen, MD
David Fish, MD, MPH
Jae Jung, MD
Elizabeth Lord, MD
Don Y. Park, MD
A. Nick Shamie, MD*
Sports Medicine
Edward Cheung, MD
Sharon Hame, MD
Kristofer Jones, MD
Thomas Kremen, MD
David McAllister, MD*
Sports Medicine & Non-Operative Orthopaedics
Calvin J. Duflaut, MD
Vanessa S. Franco, MD
Joshua Goldman, MD
Joseph P. Luffman, MD
Emily M. Miller, MD
Audra Nativi, MD
Daniel Vigil, MD*
Jessica, Zamrd, DO
Trauma Surgery
Eric Farrell, MD
Devon Jeffcutt, MD
Eric Johnson, MD*
Christopher Lee, MD
Emeritus
Andrea Cracchiolo III, MD
Jeffery Eckardt, MD
Gerald Finerman, MD
* Division Chiefs